Constraining the Atmospheric Carbon Budget: A Preliminary Assessment

I.G. Enting





Atmospheric Research

CSIRO Division of Atmospheric Research Technical Paper no. 25

Constraining the Atmospheric Carbon Budget: A Preliminary Assessment

I.G. Enting

National Library of Australia Cataloguing-in-Publication Entry

Enting, I.G. Constraining the Atmospheric Carbon Budget

Bibliography. ISBN 0 643 05249 6 (print edition) 0 643 06506 7 (electronic edition)

1. Atmospheric carbon dioxide – Measurement. 2. Carbon cycle (biogeochemistry) – Mathematical models.

I. CSIRO. Division of Atmospheric Research. II. Title. (Series: Division of Atmospheric Research Technical Paper no. 25)

551.5112

Address and contact details: CSIRO Atmospheric Research Private Bag No.1 Aspendale Victoria 3195 Australia Ph: (+61 3) 9239 4400; fax: (+61 3) 9239 4444 e-mail: chief@dar.csiro.au

CSIRO Atmospheric Research Technical Papers may be issued out of sequence.

© CSIRO Australia 1992 (electronic edition 2000).

Constraining the Atmospheric Carbon Budget: A Preliminary Assessment

I.G. Enting CSIRO Atmospheric Research Private Bag 1, Aspendale, Vic 3195, Australia

1. Introduction

Direct observations show that the concentration of CO_2 in the atmosphere has been increasing over the last three decades (from about 315 ppmv in 1959 to 350 ppmv in 1990) and that the rate of increase has grown. Measurements of gas trapped in bubbles in polar ice show that the concentrations were around 280 ppmv at the beginning of the last century and have grown at a generally increasing rate over the last two hundred years. The current rate of increase of atmospheric CO_2 represents about half of the rate at which carbon is released due to the use of fossil fuels.

While it is generally assumed that the world's oceans represent the sink for much of the fossil carbon that leaves the atmosphere, the role of the terrestrial biota in the atmospheric carbon budget is the subject of considerable uncertainty. In particular, the study by Tans et al. (1990) raised the possibility that the oceanic uptake of CO_2 might be only half as much as determined by conventional carbon cycle modelling, implying very large biotic uptakes.

Emissions	Gt C y^{-1}
Fossil carbon release	5.4 ± 0.5
Deforestation and land use change	1.6 ± 1.0
Accumulation in atmosphere	3.4 ± 0.2
Uptake by ocean	2.0 ± 0.8
Net imbalance	1.6 ± 1.4

Table 1: Atmospheric carbon budget for 1980–90, from IPCC (1990, p13).

Table 1 presents the estimated atmospheric carbon budget for the period 1980–90 as presented in the IPCC (1990) report. Given recent developments in carbon cycle studies it seems appropriate to refine this presentation in a number of ways:

- i. Explicitly recognising the possible role of extra CO₂-induced growth of the terrestrial biota, even if direct estimates have much greater uncertainties than those arising from atmospheric carbon budgeting.
- **ii.** A greater recognition of the fact that the budget is changing in response to anthropogenic inputs. The changes over the decade 1980–1990 have been comparable to the uncertainties

in some components of the budget and a comprehensive analysis needs to separate actual uncertainties from secular changes.

- iii. A recognition that some of the doubts raised by Tans et al. (1990) appear to have been resolved.
- iv. A more systematic treatment of the range of uncertainty.

It must be emphasised that an understanding of the current atmospheric carbon budget confers only a limited predictive power for estimating the future atmospheric CO_2 levels resulting from various possible anthropogenic emissions. Among the possible changes that will be neglected are:

- Direct effects on CO₂ uptake from changes in ocean circulation that result from either natural or anthropogenic climate change;
- Indirect effects of such circulation changes due to changed nutrient supply to the marine biota.
- Direct effects of climatic change and changes UV-B on the marine and terrestrial biota.

Concerns such as these are noted in Section 1.2.7 (and subsections thereof) of IPCC (1990).

The present report has the following aims:

- i. Presenting the methodology for using linear programming techniques to analyse atmospheric constituent budgets, for use by those working on different aspects of the global carbon cycle.
- **ii.** Presenting a specific set of estimates of the limits on the atmospheric carbon budget, within the present state of knowledge of global constraints.
- **iii.** Presenting preliminary analyses of the limits on the atmospheric carbon budget based on regional carbon constraints, incorporating the results of atmospheric transport modelling.

The estimated carbon budgets will be presented at three levels of detail. Each step represents the inclusion of additional indirect information, thus tightening the constraints, but it also involves additional assumptions about the correctness of the interpretations placed on such indirect information.

The three levels of detail are:

i. Direct estimates of each of the main contributions to the atmospheric carbon budget. These are described in Section 2 and used in Section 3 to illustrate the linear programming formalism.

- **ii.** The use of ocean models to estimate the oceanic uptake of CO₂. The various approaches are discussed in Section 4.
- iii. The use of regional carbon budgets derived from analysing spatial distributions of CO_2 using atmospheric transport models. The possibilities and data requirements are discussed in Section 5.

In order to take into account the change in the budget over time, we consider three different periods:

1978 As a first point for comparisons.

1988 As a second point to show the changes in the atmospheric carbon budget with time.

1980-5 For comparison with earlier work, especially that based on spatial distributions of CO₂.

The years 1978 and 1988 are used as reference points for the comparisons that illustrate the extent of changes in the atmospheric carbon budget. These are nominal time points for comparison. Since instantaneous rates of change can not be reliably determined from observational data, the figures we use are averages, generally over 2 or more years, normalised to the middle of the reference years 1978 and 1988. The period 1980–5 is the most suitable for looking at the regional budgets because of the greater quantity of data available. Records from before this time are subject to unknown offsets relative to each other because of lack of adequate intercalibration. In many cases records from later periods are yet to be published. However the interpretation of the atmospheric budgets for the period 1980–1985 is complicated by a number of anomalies:

- i. The fossil fuel emissions ceased to grow over this period.
- ii. The 1982–3 ENSO event had a large effect on CO_2 (see for example Gaudry et al., 1987).
- **iii.** The interhemispheric gradient for this period seems anomalously low, possibly as a consequence of one or both of the other two effects noted.

The outline of the remainder of this report is as follows. Section 2 reviews the global constraints on the atmospheric carbon budget. Section 3 describes the linear programming formalism for analysing the carbon budget and applies the formalism using the constraints from Section 2 as the first example. Section 4 gives a more detailed discussion of model-based constraints on the oceanic uptake of CO_2 , reviewing various modelling approaches. The section concludes with an analysis of how, subject to the validity of the model, model-based constraints can constrain the atmospheric carbon budget more tightly than direct estimates alone. Section 5 reviews the role of regional constraints based on atmospheric transport modelling. The results quoted from regional budgeting are purely illustrative because of the simplified nature of the transport model and the ad hoc nature of the error analysis.

2. Global Constraints

For this analysis we regard the atmospheric budget at the specified times as a balance of the 5 components, each denoted by a 6-letter identifier as used in our computer programs:

INCRES This is the rate of increase of carbon (as CO_2) in the atmosphere.

FOSSIL This is the rate of input of fossil carbon

OCEANS This is the net uptake of carbon by the oceans.

CLEARS This is the net carbon input into the atmosphere resulting from land-use changes.

FERTIL This is the amount of extra net biotic growth due to the fertilising effect of CO_2 and any other anthropogenic influences.

Each of these components is expressed in units of Gt C y^{-1} . The following subsections describe the prior constraints that can be placed on each of the components.

2a. The atmospheric increase

In order to define the rate of increase of CO_2 in the atmosphere, consistency of calibration in the observations in extremely important. The primary source of data that we use is the Scripps programs for Mauna Loa and the South Pole (Keeling et al., 1989a). The annual means for Mauna Loa and the South Pole are used, subject to limitations from missing South Pole data.

For '1978' the difference in the 1977 and 1979 annual means implied growth rates of 1.48 ppmv y⁻¹ for Mauna Loa and 1.49 ppmv y⁻¹ for the South Pole. The 1976 to 1980 difference for Mauna Loa implied 1.64 ppmv per year. We take the '1978' value as 1.5 ± 0.1 ppmv y⁻¹ or 3.18 ± 0.20 Gt C y⁻¹. The 1980 to 1985 difference for Mauna Loa implies a mean growth of 1.4 ppmv y⁻¹, while the 1979 to 1986 difference for the South Pole implies 1.48 ppmv y⁻¹. We take the rate as 1.45 ± 0.05 ppmv y⁻¹ or 3.08 ± 0.10 Gt C y⁻¹. The range for 1980–1985 is taken as half of the '1978' range because the change is being determined relative to a longer time base.

The '1988' value was estimated from the Scripps South Pole and Mauna Loa data, as published in CDIAC (1991). The respective mean growth rates from these records were 1.98 and 1.73 ppmv y^{-1} for 1987–1989 and 1.74 and 1.57 ppmv y^{-1} for 1986–1990. We take the value 1.76 \pm 0.20 ppmv y^{-1} which gives a range of 3.32 to 4.16 Gt C y^{-1} . The larger range compared to the '1978' value reflects a possible ENSO influence.

2b. Fossil carbon release

Estimates of the global annual releases of CO_2 from fossil carbon (including cement production) have been compiled by Keeling (1973), Rotty (1987) (and references therein) with recent estimates being collected in CDIAC (1990). The current methodology has been described by Marland and Rotty (1984). G. Marland presented an estimate of 5.967 Gt for 1989 at the NATO Advanced Study Institute (Il Ciocco, Italy, September, 1991). For the purposes of the present analysis we use 5.14 Gt C y⁻¹ for 1978 and 5.86 Gt C y⁻¹ for 1988, representing the averages (weighted as $(\frac{1}{4}, \frac{1}{2}, \frac{1}{4})$) for 1977–1979 and 1987–1989 respectively. The figure 5.20 Gt C y⁻¹ is used as the average for the period 1980–1985.

The uncertainty ranges applied to these estimates are +10% and -5% in the reference case. Some examples given below explore the consequences of reducing the range to +5% to -2.5%.

2c. Ocean uptake

Currently, direct measurements of CO_2 fluxes over the ocean, using gradient techniques, are possible only in favourable conditions. Thus no global (or even regional) coverage has been achieved in this way.

The most direct estimates of air-sea flux of CO_2 are those that combine measurements of the CO_2 partial pressure difference (Δp_{CO2}) with a gas-exchange coefficient to produce a CO_2 flux.

Such calculations are subject to three important limitations:

- The lack of $p_{\rm CO2}$ data for the Southern Hemisphere;
- Questions about the representativeness of Northern Hemisphere $p_{\rm CO2}$ data;
- The discrepancy between the air-sea gas exchange coefficients measured directly and those estimated using the global uptake of ¹⁴C from weapons tests.

The most comprehensive attempt at direct calculation of oceanic CO_2 uptake was that of Tans et al. (1990). They concluded that the oceanic fluxes were (with uptakes negative) -0.59 Gt C y⁻¹ north of 15°N, 1.62 Gt C y⁻¹ from 15°N to 15°S, and -2.39 Gt C y⁻¹ south of 15°S. These fluxes were calculated using monthly mean wind speeds with an empirical transfer coefficient E (in moles of $CO_2 \text{ m}^{-2} \text{ yr}^{-1} \mu \text{atm}^{-1}$) expressed as a function of windspeed W (in m s⁻¹) as

$$E = 0.016(W - 3)$$
 for $W > 3$ (1)

$$E = 0 \qquad \text{for } W \le 3 \tag{2}$$

Tans et al. indicated that use of the alternative transfer coefficients of Liss and Merlivet (1986) would lead to fluxes half of those quoted, implying ranges (-0.59 to -0.30), (1.62 to 0.81) and

(-2.39 to -1.20) for their 'northern', 'tropical' and 'southern' regions. In order to convert these ranges to the latitudinal division used below with a 'tropical' range of 12° N to 12° S, we simply scale the tropical component and add the difference equally to the other two regions, giving (-0.43 to -0.22), (1.30 to 0.65) and (-2.23 to -1.12). Tans et al. suggested that sampling errors could lead to as much as a 1 ppmv bias implying flux errors of up to 0.07 Gt C y⁻¹ for northern oceans and up to 0.15 Gt C y⁻¹ for southern oceans.

The systematic major adjustments to these ranges are the corrections noted by Sarmiento and Sundquist (1992); the effect of carbon transport by rivers and the effect of 'skin-temperature' on determining air-sea gas exchange. Sarmiento and Sundquist estimate that the carbon flux through river is adding 0.2 to 0.6 Gt C y⁻¹ to northern oceans by a transport additional to the surface fluxes calculated by Tans et al., thus changing the 'northern' range to (-1.03 to -0.42). They also suggest that the air-sea exchange will be determined by the temperature of a thin layer at the ocean surface whereas measurements of ocean p_{CO2} are through equilibration of water from a little below the ocean surface. They suggest that this could increase the calculated global uptake by 0.3 to 1.1 Gt C y⁻¹ assuming gas exchange rates consistent with oceanic uptake of ¹⁴C with about half this correction applying if the lower gas exchange coefficients were used. This leads to a set of flux ranges of (-1.43 to -0.52), (1.00 to 0.55) and (-2.63 to -1.22).

Apart from the sampling errors, most of the corrections have their limits set by the range of uncertainty in the gas exchange coefficient and so the ranges for the three regions do not represent independent variations. It is not consistent to combine the maximum tropical release with the minima of higher latitude uptake or vice versa. The global range corresponds to that obtained from summing the limits of the three ranges to give -2.63 to -1.19 Gt C y⁻¹.

Our final ranges are obtained by applying the 'sampling' error of ± 0.3 Gt C y⁻¹ globally (as suggested by Tans et al., 1990), leading to a global range of -2.93 to -0.89 Gt C y⁻¹. Applying the 'sampling' error on a regional basis (as 0.07, 0.08, 0.15 Gt C y⁻¹) gives the set of flux ranges (-1.50 to -0.48), (1.08 to 0.51) and (-2.78 to -1.15).

Of the global range of uncertainty (over 2 Gt C y^{-1}), roughly equal contributions come from: uncertainty in the air-sea gas exchange relation, sampling limitations for p_{CO2} , and the systematic corrections for river carbon and skin-temperature effects. Therefore no single improvement will lead to major reductions in the uncertainty over the global net oceanic uptake.

2d. CO₂ release from land-use changes

Houghton et al. (1983) presented a set of calculations of net CO_2 emissions resulting from land use changes for the period 1860–1980. The effects included carbon release from forest clearing either for agriculture or replanting and then the effect of regrowth from re-aforestation, together with other consequent changes, particularly those involving soil carbon.

Recent work by Houghton (personal communication) has produced a pair of estimates of the time history of release. These give the ranges 0.990 to 1.616 Gt C y^{-1} for 1978, 1.194 to 1.991

CSIRO Division of Atmospheric Research Technical Paper no. 25

Gt C y^{-1} for 1980–1985 and 1.594 to 2.550 for 1988. These estimates can probably be regarded as spanning a plausible range rather than the full range of uncertainty. Houghton et al. (1987) typically use a $\pm 50\%$ uncertainty around their best estimates. In order to produce a comparable range we have taken a lower limit of 80% of the low estimate and an upper limit of 130% of the higher estimate, giving ranges 0.79 to 2.10 Gt C y^{-1} , 0.96 to 2.56 Gt C y^{-1} and 1.28 to 3.32 Gt C y^{-1} for 1978, 1980–1985 and 1988 respectively.

The spatial distribution was discussed by Houghton et al. (1987); the division was 0.028 Gt C y^{-1} for Australia and New Zealand, 1.659 for tropical sources and 0.105 for northern sources, all applicable to 1980. For the present we neglect the fact that a small part of the tropical source (e.g. for Uganda) lies outside the range of 12°S to 12°N used in the regional study below. For the northern and southern regions we apply the $\pm 50\%$ ranges to give 0.052 to 0.157 Gt C y^{-1} and 0.014 to 0.042 Gt C y^{-1} respectively. We then subtract these from the global range to get the 'tropical' range of 0.90 to 2.36 Gt C y^{-1} .

2e. CO₂ uptake from fertilisation effects

The estimates obtained by Houghton et al. (1983) are of the carbon fluxes from the terrestrial biota occurring as a result of land-use changes. There will be a number of other effects that lead to biotic fluxes even in the absence of land-use changes.

- Increased growth due to the fertilising effect of additional atmospheric CO₂;
- Increased growth due to the fertilising effect of additional anthropogenic nitrogen compounds;
- Decreased carbon stored following damage to biota from other pollution;

Of these effects, the possibility of CO_2 -induced growth has attracted most attention. Among the various approaches to quantifying (or at least identifying) this effect have been:

- i. Global modelling, using extrapolations of growth factors from laboratory studies (e.g. Kohlmaier et al., 1987);
- ii. Interpretation of changes in the seasonal cycle of CO_2 in terms of biotic growth;
- **iii.** Searches for increased growth in tree-rings.

Interpretation of changes in the seasonal cycle of CO_2 in terms of increased biotic uptake is subject to a number of serious problems:

• A change in amplitude of the seasonal exchange does not in itself imply a change in net carbon storage and certainly provides little basis for quantifying the integrated global carbon storage.

- The change in amplitude is not observed as a coherent signal over all observational records.
- The change in amplitude at Mauna Loa seems to have occurred primarily between 1975 and 1980 with no consistent trend either before or after this period (see for example Enting and Manning, 1989). This weakens the claim for an association between amplitude changes and CO₂ changes.
- Contrary to claims by Idso (1987), the correlation structure that Enting (1987) interpreted as a characteristic biotic signal does not imply that the same cause applies to the longerterm change in amplitude. Furthermore a refined analysis (Enting and Manning, 1989) shows that the correlation is confined to the period prior to the increase over 1975–1980. Subsequent analysis of longer records by K. Thoning (personal communication) suggested that the correlation structure was reversed after 1980. Thus the Enting (1987) analysis does not constitute detection of a global-scale fertilisation effect.

The possibilities for detecting a CO_2 -fertilisation signal in tree-ring records are severely restricted by the difficulties of separating such a signal from climatic signals, although La March et al. (1984) claimed to have detected a fertilisation signal. In any case, detection of a signal in a single species (at a climatically stressed site) provides an insufficient basis for quantifying the global carbon storage.

For the present study we adopt model-based results as giving the most direct estimates available and in particular we use the results of Kohlmaier et al. (1987). Their main result is an estimated uptake rate of 1.4 ± 0.7 Gt C y⁻¹ for 1982. This represents the combination of 0.7 ± 0.3 Gt C y⁻¹ stored in living biomass, 0.07 ± 0.03 Gt C y⁻¹ stored in litter and 0.60 ± 0.35 Gt C y⁻¹ stored in humus. On a 4-biome division, the storage in the living component was quoted as 0.09 to 0.20 Gt C y⁻¹ in grasslands, 0.16 to 0.36 Gt C y⁻¹ in tropical forests, 0.13 to 0.33 Gt C y⁻¹ in temperate and boreal forests and very small amounts in agricultural systems. (The slightly wider range on the total apparently represents a degree of cautiousness on the part of the authors).

In order to estimate the way in which the results from Kohlmaier et al. (1987) can be extended to describe the variation in time and space, a related model developed by P. Polglase of Melbourne University has been used. For a β -factor of 0.2 the model predicted a biotic uptake increasing from 0.48 to 0.66 Gt C y⁻¹ over 1975–1985. For $\beta = 0.5$ the increase was from 1.21 to 1.65 Gt C y⁻¹. Therefore to extend the Kohlmaier et al. 1982 value to 1978 and 1988, we apply shifts of -0.1 and +0.1 to the lower bound and -0.3 and +0.3 to the upper bound, having scaled-up the upper rate-of-change to bring the rates into agreement with the upper estimate of Kohlmaier et al. (1987). The Polglase model was also used to investigate the distribution of uptakes between biomes. For both the upper and lower β values tropical forests were found to contribute close to a third of the total uptake. This is smaller than the 50% fraction applying to the living biomass because of the much smaller component of soil carbon storage calculated for tropical forests.

3. Methodology

3a. General

The analysis of sets of constraints such as those developed in the previous section is handled using a *linear programming* approach.

Note that the term *programming* is used in this context with the meaning (standard in many branches of mathematics) of 'scheduling' and does not specifically refer to computer programming.

The aim of the formalism is to give a way of combining the estimates of the various contributions to the atmospheric carbon budget, together with their uncertainties, in order to give a consistent picture of the final uncertainty remaining after all of the various constraints have been incorporated.

Given a decomposition of the atmospheric carbon budget into a set of components and the formulation of a set of (linear) constraints applying to these components, the linear programming formalism can be used in several ways to explore the range of possible solutions. The simplest question to ask is that of the allowed range of one or more of the components of the budget. However the formalism also provides the capability for exploring the allowed range of joint variability of a set of two or more variables. We produce what we call 'mutual-consistency-regions' (MCRs) which we define to be the set of contributions consistent with all the constraints. For N components, the MCR will be an N-dimensional polyhedron. We also consider the range of consistency of subsets of variables. The MCR of such a set of M components will be the projection of the N-dimensional region onto the M-dimensional space of components considered.

3b. The mathematical formalism

The mathematical expression of the linear programming problem is defined in terms of a set of N variables, x_j for j = 1 to N. In the present study these correspond to INCRES, FOSSIL etc. (The actual ordering of the variables is generally irrelevant and for the programs used here is taken from the input file defining the code names.) The x_j are required to be inherently positive. The constraints $x_j \ge 0$ are implicit in the linear programming formalism and need not be entered explicitly. Any quantity that can be of either sign must be re-expressed in terms of possible positive and negative contributions:

$$Y = Y_POS - Y_NEG$$

with Y_POS and Y_NEG assigned to different x_j . This issue can become important when considering detailed regional carbon budgets.

The variables x_k are regarded as being subject to various linear constraints:

$$\sum_{k} a_{jk} x_{k} \leq e_{j}$$
$$\sum_{k} b_{jk} x_{k} \geq f_{j}$$
$$\sum_{k} c_{jk} x_{k} = g_{j}$$

The linear programming problem is to find the maximum (or minimum) of some linear combination of the x_j , i.e. a sum such as $\Psi = \sum_j \psi_j x_j$.

Implementations of algorithms for solving this linear programming problem can differ in various sign conventions adopted. The first choice is whether a particular procedure maximises or minimises Ψ . Changing the sign of all the ψ_j converts a maximisation problem into a minimisation problem without loss of generality. A second choice is usually made to force all the inequalities into a standard form. Inequalities \leq and \geq are reversed by reversing the signs of all coefficients. This can be used either to make all the inequalities of the one direction or all the right-hand sides of one sign.

In applying the linear programming formalism to studying the atmospheric carbon budget, the first thing that can be done is simply to look at the extreme values of each of the components. This, is simply a matter of setting $\psi_j = \pm 1$ for one particular j value, with all other ψ_j zero. Greater insight into the range of uncertainty in the atmospheric carbon budget can be obtained by looking at the jointly allowed values of two distinct components. This can be done by maximising $a_j x_j + a_k x_k$ for the desired choices of j and k for a range of a_j , a_k defining different directions in the x_j - x_k plane. Appropriate choices of a_j and a_k will determine the vertices of the polygonal bound of the MCR in the x_j - x_k plane and this region can then be plotted. Similarly more complex combinations can be explored but the results are harder to represent visually. In order to explore the joint variability of 3 quantities x_j , x_k and x_m we choose a set of allowed values of x_m and determine cross-sections of the 3-variable MCR for various fixed x_m by calculating 2-variable MCRs with the additional constraint of fixed x_m .

3c. The combined uncertainties

In order to illustrate the formalism described above, we apply it to the simplest case of a global description of the atmospheric carbon budget, using the constraints developed in Section 2 above.

As outlined in the introduction we present estimated atmospheric carbon budgets at two distinct times, 1978 and 1988, recognising that these refer to running averages around these times rather than budgets for the single years. In addition we consider the period 1980–1985, subject to the caveats concerning the apparently anomalous behaviour of the cycle.

Table 2 displays the constraints described in Section 2 above with '78', '88' and '83' representing three alternative sets of 'right-hand-sides' applying for 1978, 1988 and 1980–5 respectively.

INCRES	FOSSIL	OCEANS	CLEARS	FERTIL		'78'	'88'	'83'
-1.0	1.0	-1.0	1.0	-1.0	=	0.0	0.0	0.0
1.0	0.0	0.0	0.0	0.0	\geq	2.98	3.32	2.98
1.0	0.0	0.0	0.0	0.0	\leq	3.38	4.16	3.18
0.0	1.0	0.0	0.0	0.0	\geq	4.88	5.57	4.94
0.0	1.0	0.0	0.0	0.0	\leq	5.65	6.45	5.72
0.0	0.0	1.0	0.0	0.0	\geq	0.80	0.98	0.89
0.0	0.0	1.0	0.0	0.0	\leq	2.64	3.32	2.93
0.0	0.0	0.0	1.0	0.0	\geq	0.79	1.28	0.96
0.0	0.0	0.0	1.0	0.0	\leq	2.10	3.32	2.56
0.0	0.0	0.0	0.0	1.0	\geq	0.60	0.80	0.70
0.0	0.0	0.0	0.0	1.0	\leq	1.80	2.40	2.10

Table 2: Constraints based on direct estimates of components of the atmospheric carbon budget. Units are Gt C y^{-1} . Each line in the body of the table represents a distinct constraint for each of the 3 times considered.

It is found that combining the information does not reduce the initial range of uncertainty for any of the variables in any of the cases. This was to be expected since there is only one constraint (a balanced atmospheric budget) linking variables that are otherwise independent. The effects of the constraint become apparent in the plots of the mutual consistency regions (MCRs) of various sets of variables. Figure 1 shows the allowed joint variation of OCEANS and CLEARS. It shows that the combination of low oceanic uptake and high rates of clearing cannot both occur together. Figure 2 shows the allowed joint variation of OCEANS and FERTIL. In this case the combination of low oceanic uptake and low fertilisation uptake are seen to be excluded.

It is Figure 3 that gives a better idea of the inter-relation of the constraints. It is based on a modification of the standard case. The fossil fuel range is reduced to (-2.5% to +5%). Figure 3 shows the combined ranges of CLEARS and FERTIL for various fixed values of OCEANS, i.e. the figure shows cross-sections if the 3-variable MCR for OCEANS, CLEARS and FERTIL. From the figure it becomes clear that for any fixed value of oceanic uptake, the net biotic source (CLEARS – FERTIL) is constrained. The allowed values of CLEARS and FERTIL form a diagonal band. The length of the band is set by the prior constraints on the two quantities. The width of the band (as measured parallel to either axis) is the sum of the ranges of INCRES and FOSSIL.

Finally, in Figure 4 we show the effect on the MCRs of the secular change in the global carbon system. The figure shows the 2 MCRs for the '1978' and '1988' cases. Comparisons with the corresponding region for the 1980–1985 case shown in Figure 1 shows an increase in the absolute range of uncertainty over time. For most components, the uncertainty grows with the size of the source or sink. The displacement between the two regions shown in Figure 4 is a small but significant part of the allowed range of variability, even with the large uncertainties that remain in budgets based solely on direct estimates. When additional information is supplied to reduce these ranges, the secular changes can become comparable to the range of uncertainty.



Figure 1: Joint variation of OCEANS and CLEARS given the constraints from Table 2 for the 1980–5 case. The region within the polygon corresponds to the set of mutually consistent values for the net oceanic uptake and the net rate of carbon release from land-use changes.



Figure 2: Joint variation of OCEANS and FERTIL given the constraints from Table 2 for the 1980–5 case.



Figure 3: Joint variation of FERTIL and CLEARS for various fixed values of OCEANS: (a) 1.2 Gt C y^{-1} , (b) 2.0 Gt C y^{-1} and (c) 2.8 Gt C y^{-1} , given the constraints from Table 2 for the 1980–5 case except that the range of fossil fuel release is taken as 5.07 to 5.46 Gt C y^{-1} . The three polygons are cross sections of the 3-dimensional region of allowed variability of OCEANS, CLEARS and FERTIL.



Figure 4: Joint variation of OCEANS and CLEARS given the constraints from Table 2 for the 1978 and 1988 cases (lower left and upper right polygons respectively).

4. Model-based Constraints on Oceanic Uptake

4a. General

As an alternative to the use of direct information about the oceanic uptake of CO_2 we can follow the 'traditional' approach from carbon cycle modelling and calculate the oceanic CO_2 uptake using a model. Even in the context of a 'perfect' model, the present oceanic uptake of CO_2 will depend on the history of past changes. The relationship can be conveniently expressed using the response function formalism introduced by Oeschger and Heimann (1983).

Given the assumption of stationarity, a linear treatment gives the relation between the net terrestrial source, S(t), and the concentration, C(t), as:

$$C(t) - C(t_0) = Q(t) = \int_{t_0}^t R(t - t')S(t') dt'$$
(3)

where R(t) is a response function that gives the amount of CO_2 remaining in the atmosphere at a time t after a unit input. This relation has a formal inverse:

$$S(t) = \dot{Q}(t)/R(0) - Q(t)\dot{R}(0)/R(0)^2 - \int_{t_0}^t K(t - t')Q(t')\,dt'$$
(4)

The inversion kernel K(.) is related to R(.) (Enting and Mansbridge, 1987b). If R is approximated as a sum of N exponentials then K can be readily derived as a sum of N-1 exponentials. The oceanic uptake $\Phi(t)$ can be written as:

$$\Phi(t) = S(t) - \dot{C}(t) / R(0) = -Q(t)\dot{R}(0) / R(0)^2 - \int_{t_0}^t K(t - t') Q(t') dt'$$
(5)

This relation shows that the oceanic uptake for a particular time t is expressed in terms of the history of concentration change without having to make explicit reference to the history of sources that produced that change.

With the availability of measurements of past CO_2 concentrations from bubbles in polar ice (e.g. Neftel, 1985) it has been possible to determine the current oceanic carbon uptake by either:

- Using a full carbon cycle model with the atmospheric concentration constrained to follow the ice-core record;
- Using the response function representation of such a model with the atmospheric concentration constrained to follow the ice-core record;
- Using the inversion relation above with the kernel K(.) derived from the model response function.

The two terms in the expression for Φ have opposite signs (K positive and $\dot{R}(0)$ negative) and so in principle Φ could show variations more rapid than C(t). However the effect of steadily increasing C(t) will be steadily increasing oceanic uptake, Φ . To quantify this we use the results of the modelling study reported by Enting (1991) (which formed the basis of some of the calculations reported in the IPCC report). The oceanic uptake in this modified version of the box-diffusion model (with the parameters taken from Broecker et al., 1980) was 1.6 Gt C y⁻¹ for 1978, 2.0 Gt C y⁻¹ for 1988 and 1.8 Gt C y⁻¹ averaged over 1980–1985. For the purposes of the present analysis we use factors of -10% and +10% to rescale direct estimates for 1980–1985 to be applicable to 1978 and 1988 respectively.

In spite of the wide variety of models, several important common features occur, distinguishing modelling from direct estimation of CO_2 uptake. Firstly, p_{CO2} is calculated from the model equations rather than relying on the limited observational data. Secondly, the oceanic uptake over long periods is only weakly dependent on the air-sea gas exchange coefficient *E*. Finally, correction terms such as the skin effect and river carbon have their main influence on the steady-state distribution of carbon between atmosphere and ocean and have only minor second-order effects on the response to perturbations in atmospheric CO_2 . Thus the main causes of uncertainty in direct estimates (see Section 2c above) are largely bypassed by ocean modelling, but the results are conditional on the validity of the model that is used.

4b. Non-stationarity

The formalism outlined in the previous section applies to general linear stationary models. The question of linearity is not a serious issue. The main non-linearity in oceanic uptake involves the relation between carbon content and $p_{\rm CO2}$ in the ocean mixed layer. However in describing the response to additional carbon inputs, this non-linearity only becomes significant at concentrations much greater than those occuring at present.

A more important question is whether the ocean transport processes have remained unchanged in time. A related question is whether the pre-industrial distribution of CO_2 in the atmosphere and oceans represented an equilibrium consistent with present oceanic transport of whether the pre-industrial distribution reflected a different transport regime in the past.

On the time scale of 2 to 4 years the El Niño/Southern Oscillation (ENSO) phenomenon represents a non-stationary influence on atmospheric CO_2 .

One piece of evidence for a longer-scale non-stationarity within the industrial period is the study by Enting and Mansbridge (1987) and the refined analysis by Enting (1992). The result of these studies is that there is no response function for a linear steady-state ocean that can reconcile the CO_2 record from ice-cores with the release estimate obtained by summing the fossil carbon estimates from Rotty and the biotic sources estimated by Houghton et al. (1983). The study by Enting (1992) shows that including a fertilisation effect in the calculation increases the discrepancy. The solution proposed by Enting (1992) was that the ice-core record for the early part of the industrial period included an oceanic component reflecting some non-stationary behaviour, possibly a recovery from a perturbation associated with the little ice age.

The possibility of non-stationarity in the ocean mixing processes remains as a caveat to all current use of ocean models for either interpreting the atmospheric carbon budget or for predicting future CO_2 levels.

4c. Parameterised Models

Most carbon cycle modelling has made use of highly parameterised representations of the ocean. Perhaps the most widely used of such models is the 'box-diffusion model' introduced by Oeschger et al. (1975) and used (with a variety of refinements) in many subsequent studies.

Because of the 'lumped' structure of such models, the ocean mixing parameters have little direct relation to observable physical transport and so the model mixing has to be determined by calibrating the model using observed tracer distributions. Most commonly ¹⁴C is used for such calibration.

In many modelling studies, sensitivity studies have been carried out, investigating the extent to which model results are sensitive to variations in the model parameters. However, given the way that the models are calibrated, the most relevant question is the extent to which the model results depend on those parameter variations (and combinations of variations) that are consistent with the calibration data.

There have been two studies that addressed this more complicated question. Gardner and Trabalka (1985) used a 'Monte Carlo' approach, generating a large sample of pseudo-random sets of parameter variation and analysed the properties of the sub-sample that was consistent with various carbon cycle data. Enting and Pearman (1986, 1987) used numerically determined derivatives to search for the directions in parameter space having the most extreme sensitivity consistent with the calibration data. This was done within the context of a 'Bayesian' calibration that fitted both the carbon cycle data and independent estimates of the model parameters.

Recently this approach has been used on a modified version of the box-diffusion model to explore the range of possible uptakes consistent with carbon cycle data. This study was in terms of a restricted set of changable parameters, chosen to reflect those which the preliminary analytic study by Enting (1990) had suggested as most critical in determining the range of oceanic uptake consistent with the data. In particular the model was used to explore the possible role of detrital transport in removing ¹⁴C from the ocean surface so that the observed uptake of ¹⁴C from nuclear tests could be reconciled with low oceanic uptake of CO_2 .

The Bayesian calibration adjusted 6 model parameters, 2 for detrital transport (organic and carbonate detritus having different depths of re-dissolution), the eddy mixing in the ocean, the atmospheric turnover time with respect to ocean uptake, and the two parameters for ¹⁴C production (from cosmic rays and nuclear tests). The model was run in inverse mode, tracking the CO₂ record from ice cores. The model output (primarily isotopic information since atmospheric CO₂ changes were fitted exactly) was fitted to a set of 15 data items similar to that used by Enting and Pearman (1987) without the spatial disaggregation.

The range of oceanic uptake was explored by using the Lagrange multiplier formalism described by Enting (1985) and applied by Enting and Pearman (1986, 1987). It was found that a range of about ± 0.33 Gt C y⁻¹ of biotic exchange approximately doubled the sum of squares of residuals relative to the best fit. For 15 independent variables (given 6 variables fitted to 15 + 6 quantities) a doubling corresponds to a χ^2 of 30 and thus a 'confidence' value of 99%. The range of ± 0.33 Gt C y⁻¹ was adopted for the analyses presented below.

4d. Direct inversion

A somewhat different approach to the calibration of lumped models of ocean circulation seeks to obtain transport coefficients more closely linked to actual water-mass movements by inverting the distribution of a suite of chemical tracers. This approach was attempted by Bolin et al. (1983) who used carbon, oxygen, alkalinity, phosphorus and ¹⁴C to determine the transport coefficients (and biological contributions) for a 12-box representation of the global ocean.

The inversion calibration of the 12-box model leads to significant discrepancies between observed and calculated distributions of phosphorus. The CO_2 uptake calculated by the model using fossil fuel inputs corresponded to an airborne fraction of 70%. The relevance of this calculation is limited by the neglect of biotic inputs, probably implying too low a value for atmospheric CO_2 , and the fact that the very large uptake also led to lower than observed CO_2 , thus giving an underestimate of Φ . A more satisfactory approach, avoiding both these problems would be to run the model with the atmospheric CO_2 concentration history specified by the record from ice-cores.

The authors regarded the model results as limited by the resolution. Later work by the same group has used much greater resolution but has concentrated on the Atlantic Ocean (see for example Moore and Björkström, 1986).

4e. Ocean general circulation models

The use of ocean general circulation models that derive the ocean transport from a knowledge of the basic physics would be one way of removing much of the uncertainty in the oceanic uptake of CO_2 . At present there are two main limitations on doing this:

- The models require considerable computing resources, particularly given that characteristic scales of oceanic eddies are much smaller than for atmospheric eddies.
- The development of ocean GCMs is still continuing; in existing GCMs a number of parameterisations are required in order to produce acceptable circulation patterns.

In spite of these difficulties, there have been several calculations of oceanic CO_2 uptake using models based on the transport calculated by a GCM (Maier-Reimer and Hasslemann, 1987;

Bacastow and Maier-Reimer, 1990; Sarmiento, 1986; see also Sarmiento and Sundquist, 1992).

For the purposes of the present study, the various estimates suffer from a lack of comparability in time and a lack of any estimate of uncertainty apart from the obvious model-to-model differences. Maier-Reimer and Hasselmann (1987) quote an oceanic uptake of 40% of fossil releases (i.e. 2.05 Gt C y⁻¹ for 1978) while Bacastow and Maier-Reimer quote a fraction of 32% (i.e. 1.64 Gt C y⁻¹ for 1978). Sarmiento and Sundquist (1992) quote results of 1.9 Gt C y⁻¹ uptake for the period 1980–1989 and an average of 1.7 Gt C y⁻¹ for 1972–1989.

The most satisfactory way of achieving comparibility of time periods would be to take the response function R(t) for each model and derive the oceanic uptake by applying relation (5) to the CO₂ record from ice-cores. This requires a knowledge of the response function for each model. For example Maier-Reimer and Hasselmann (1987) represent the CO₂ uptake in their ocean model by

$$R(t) = 0.131 + 0.201 \exp(\frac{-t}{362.9}) + 0.321 \exp(\frac{-t}{73.6}) + 0.249 \exp(\frac{-t}{17.3}) + 0.098 \exp(\frac{-t}{1.9})$$

Given a response function R(t) and a specification of atmospheric CO₂ changes C(t), the oceanic uptake can be calculated for any time t. Thus a specification of R(t) greatly simplifies comparisons between models.

4f. Use of constraints on ocean uptake

The inclusion of model-based estimates of the global oceanic uptake of CO_2 does not change the structure of the analysis presented in Section 3c above. All that happens is that the rather broad band of direct estimates is replaced by a somewhat narrower range. Given the range of approaches to ocean carbon modelling there is no 'one-true-range' to be quoted on the basis of current knowledge. It is simply a matter that the more that is assumed about the validity of a model, the tighter the range that can be obtained, conditional on the validity of the assumptions made.

In presenting the analysis of model-based constraints, we use

$$1.47 \le \text{OCEANS} \le 2.13 \tag{6}$$

based on the Bayesian calibration of the box-diffusion model. This applies to 1980–1985; shifts of -0.2 and +0.2 give the 1978 and 1988 values.

This reduction in the range relative to the prior estimates is not sufficient to reduce the possible range of any of the other quantities to less than their respective prior ranges, apart from a minor reduction in the upper bound on CLEARS from 2.56 to 2.47 Gt C y^{-1} . Figure 5 shows the allowed joint range of CLEARS and FERTIL. Compared to the examples in section 3, excluding the upper range of oceanic uptake has precluded combinations of high deforestation and small amounts of fertilisation.



Figure 5: Joint variation of FERTIL and CLEARS, given the constraints from Table 2 for the 1980–5 case but with the range of oceanic uptake determined by Bayesian calibration of the box-diffusion model.

This results must be considered as an illustrative example, being subject to the validity of a highly simplfied model. While improvements in ocean GCM models can be expected to increase the confidence in their results, for the present all attempts to use ocean modelling to constrain the atmospheric carbon budget must be regarded as model-dependent.

5. The Role of Regional Constraints

5a. General

Over the last few decades, the spatial coverage of CO_2 -observing networks has increased considerably. The aim of these programs is to use the spatial distribution of CO_2 concentrations to deduce the spatial distribution of sources and sinks, thus constraining the possible processes. The determination of source strengths from concentration data requires the use of an atmospheric transport model.

The use of spatially disaggregated budgets is limited by the problem raised in the previous section — that of error bounds for the model calculations. None of the initial deconvolutions (Enting and Mansbridge, 1989; Tans et al., 1989; Enting and Mansbridge, 1991) included any sort of estimated range of uncertainty.

As an example of how regional constraints can be used, we consider a simple 3-region division into 'north', 'south' and 'tropical'. For comparison with earlier work, we use the division

considered by Enting and Mansbridge (1991), taking the 'tropical' region as being between 12°S and 12°N. The results of a source deduction using a transport model will be in the form of regional sources which we denote T_RELS, N_RELS and S_SINK. The various surface source processes have to be similarly divided into components that give the correct total so that:

- OCEANS = N_SEAS + S_SEAS T_SEAS
- $FOSSIL = N_FOSS + S_FOSS + T_FOSS$
- CLEARS = N_CLER + S_CLER + T_CLER
- $FERTIL = N_FERT + S_FERT + T_FERT$

(It is assumed that all the processes have regional fluxes in the same sense as the total, except for the oceans which are assumed to be a net source in the 'tropical' region and net sinks at higher latitudes.)

The regional source balances are:

- $N_RELS = N_FOSS + N_CLER N_SEAS N_FERT$
- T_RELS = T_FOSS + T_CLER + T_SEAS T_FERT
- S_SINK = S_FOSS S_CLER + S_SEAS + S_FERT

What is needed to complete this analysis is error bounds for the regional source components. A significant complication is that these error ranges will *not* be independent. One constraint that reduces the independence of the regional source components will be the relation:

• INCRES = $N_RELS + T_RELS - S_SINK$

The effect of this constraint could be included by including the equality into the linear programming formalism; it will imply upper and lower bounds on the (algebraic) sum of regional sources, arising from the range of increases compatible with the data. However such constraints should arise directly as part of the error analysis that considers the range of regional source components that are consistent with the data. Unless the error analysis of the source deduction can produce such constraints (or subsume them under stronger constraints) it can not be regarded as adequate. This question is further addressed in Section 5c below.

5b. Constraints from direct estimates

In most cases the direct estimates presented in Section 2 were obtained as sums over regional contributions and so there is no difficulty, in principle, in obtaining corresponding regional estimates, although in some cases the relevant data may be less accessible than the global totals.

- **Fossil** For fossil carbon release, we use the distribution given by Enting and Mansbridge (1991) with 3.8%, 3.5% and 92.7% of emissions from the 'south' 'tropical' and 'north' regions respectively. The range of uncertainty is again taken and -5% to +10%. As noted for illustrative purposes below, the regional estimates will contain uncertainties not present in the global totals, arising from effects such as fuel use in transport. However, these effects will be relatively small and are neglected in the present analysis.
- **Oceans** The direct estimates are taken from the work of Tans et al. (1990), modified by the correction terms from Sarmiento and Sundquist (1992) as described in Section 2. Since one third to one half of the uncertainty comes from uncertainty in the gas exchange formalism, assuming a consistent treatment gives a constraint on the global total that is tighter than the combination of the individual regional ranges.
- **Clearing** As discussed in Section 2d above, Houghton et al. (1987) give a detailed description of the spatial distribution of net CO_2 emissions resulting from land-use changes.
- **Fertilisation** The estimates from Kohlmaier et al. (1987) as quoted in section 2e above are converted to regional estimates by (i) taking their estimates for the living component of extra uptake in tropical forests (ii) using the Polglase model to suggest a very small contribution from litter and soil in tropical forests (iii) using the zonal estimates of NPP assembled by Pearman and Hyson (1986) to give a 2:1 ratio of NPP in our northern and southern regions.

On the basis of this information we take the living component as having the range 0.14 to 0.43 Gt C y^{-1} in the north, 0.16 to 0.36 Gt C y^{-1} in the tropics and 0.08 to 0.21 Gt C y^{-1} in the south. The non-living component is assigned the ranges 0.2 to 0.6, 0.0 to 0.2 and 0.1 to 0.3 (all Gt C y^{-1}) for north, tropics and south respectively.

5c. Constraints from atmospheric modelling

Modelling of the spatial distribution of atmospheric CO_2 can be performed in two main ways known as forward modelling and inverse modelling. In forward modelling, surface fluxes are modelled and an atmospheric transport model is used to calculate the resulting CO_2 concentrations. The fluxes can be adjusted to bring the calculated concentrations into better agreement with observations. In inverse modelling, the model is run with the surface concentrations forced to follow a specified concentration field. The surface sources are then calculated by mass balance. Examples of forward modelling studies of the spatial distribution are Pearman and Hyson, 1980, 1986; Fung et al., 1983; Pearman et al., 1983; Keeling et al., 1989b; Tans et al., 1990). Examples of inverse modelling are Enting and Mansbridge, 1989, 1991; Tans et al., 1989).

As noted above, there has been relatively little attention paid to the question of error estimates for sources derived from atmospheric transport modelling. In order to explore the extent to which such modelling can reduce uncertainties in the atmospheric carbon budget we present some preliminary analyses based on a purely diffusive representation of atmospheric transport. Such modelling has been used to explore the degree of ill-conditioning in the problem of deducing source-strengths from atmospheric concentration data (Newsam and Enting, 1988; Enting and Newsam, 1990).

The analysis considers the question of the surface source strength of CO_2 , integrated over various latitude bands. It is based on fitting observed annual mean CO_2 concentrations (as a weighted sum of Legendre polynomials), deducing the source strength implied by the concentrations and then adjusting the concentration fit so as to determine the maximum and minimum of the latitudinally integrated source, subject to the concentration curve having a specified level of agreement with the data.

There is, as yet, no good statistical basis on which to deduce error ranges for this problem. A preliminary discussion is given by Enting and Pearman (1992) where it is emphasised that it is vital to characterise the 'noise' processes that influence the observational data.

The ranges used in the present example are based on fitting 19 annual mean concentrations as given by Tans et al. (1990). (The data from Shemya and the Azores were omitted because they appeared too anomalous to be fitted with a zonally-averaged representation.). The remaining data points were fitted using Legendre polynomials up to k = 11 with the requirement that the fit agree with each data item to within 0.4 ppmv. An additional smoothness constraint was applied, bounding the magnitude of the coefficients of the Legendre polynomials P_k by 10/k. The trend was taken as 1.45 ± 0.05 ppmy y⁻¹. This led to the ranges (-1.96 to -0.51), (0.09 to 1.28) and (3.24 to 3.89) for southern, tropical and northern regions respectively. Changing the order of the fit to k = 10 reduced the southern range to -1.87 to -0.63); reducing k to 9 made it impossible to fit the data to the required precision.

As noted above, these various ranges are not independent. However the constraints on the sums of pairs of sources is essentially given by combining the constraint on the third regional component with the requirement that the total match the atmospheric increase. Other aspects of the covariance remain to be explored.

5d. Combining regional constraints

As noted above, the example that we present here follows Enting and Mansbridge (1991) in considering a 'tropical' region of 12°S to 12°N. The additional working variables that we use are N_RELS, N_FOSS, N_CLER, N_SEAS, N_FERT, T_RELS, T_FOSS, T_CLER, T_SEAS, T_FERT, S_SINK, S_FOSS, S_CLER, S_SEAS, S_FERT and INCRES. The equalities expressing the global variables OCEANS, FOSSIL, CLEARS and FERTIL as sums of their regional contributions could be used to remove these variables from the calculation, but it is convenient to retain them in the calculation in order to explore their possible ranges. Very few of the direct estimates will be more precise on a global scale than would be implied by the combination of regional constraints. In cases where a more precise global constraint is available (e.g. for fossil fuel where transport use gives a regional uncertainty that is absent from the global totals) it would still be possible to avoid using a variable for the global total by expressing the more precise global bound in terms of the sum of regional contributions.

The regional budget is only analysed for the period 1980–1985 because of the lack of wellintercalibrated data from earlier periods and the fact that much of the more recent observations are either unpublished or else only published in provisional form. The constraints based on direct estimates are:

$2.98 \leq \text{INCRES} \leq 3.18$	(7)
$2.98 \leq \text{INCRES} \leq 3.18$	(I)

$$0.19 \le \mathbf{S_FOSS} \le 0.22 \tag{8a}$$

$$0.17 \le \mathbf{T_FOSS} \le 0.20 \tag{8b}$$

$$4.58 \le \mathbf{N_FOSS} \le 5.30 \tag{8c}$$

$$0.01 \le \mathbf{S_CLER} \le 0.04 \tag{9a}$$

$$0.90 \le \mathsf{T_CLER} \le 2.36 \tag{9b}$$

 $0.05 \le \mathbf{N_CLER} \le 0.16 \tag{9c}$

$$0.18 \le \mathbf{S_FERT} \le 0.51 \tag{10a}$$

- $0.16 \le \mathsf{T_FERT} \le 0.56 \tag{10b}$
- $0.36 \le \mathbf{N_FERT} \le 1.03 \tag{10c}$

$$1.15 \le \mathbf{S_SEAS} \le 2.78 \tag{11a}$$

$$0.51 \le \mathbf{T_SEAS} \le 1.08 \tag{11b}$$

$$0.48 \le \mathbf{N_SEAS} \le 1.50 \tag{11c}$$

with a constraint on the total oceanic uptake of:

$$0.89 \le S_SEAS - T_SEAS + N_SEAS \le 2.93$$
(12)

The constraints based on atmospheric transport are provisionally set at

$$0.51 \le \mathbf{S_SINK} \le 1.96 \tag{13a}$$

$$0.09 \le \mathsf{T_RELS} \le 1.28 \tag{13b}$$

$$3.24 \le \mathbf{N_RELS} \le 3.89 \tag{13c}$$

The equalities defining relations between the components are:

$$N_{RELS} = N_{FOSS} + N_{CLER} - N_{SEAS} - N_{FERT}$$
(14*a*)

$$T_RELS = T_FOSS + T_CLER + T_SEAS - T_FERT$$
(14b)

$$S_SINK = -S_FOSS - S_CLER + S_SEAS + N_FERT$$
(14c)

$$INCRES = N_RELS + T_RELS - S_SINK$$
(15)

The linear programming analysis was applied to these constraints (48 constraints on 20 variables). Many of the quantities were not constrained beyond the constraints listed above. Cases in which the derived constraints differed from the initial direct estimates were:

FERTIL > 0.84	c.f. 0.70 prior limit	(16a)
CLEARS < 1.36	c.f. 2.56 prior limit	(16b)
$T_SEAS < 0.77$	c.f. 1.08 prior limit	(16c)
S_SEAS < 2.04	c.f. 2.78 prior limit	(16d)
T_CLER < 1.16	c.f. 2.36 prior limit	(16e)
T_RELS > 1.02	c.f. 0.09 prior limit	(16f)
S_SINK > 1.08	c.f. 0.51 prior limit	(16g)

The reduction in total clearing reflects the constraint on the tropical component. Figure 6 illustrates this. In contrast, the reduction in the southern ocean sink has not had a corresponding effect on the ocean total. This reflects the fact that the oceanic estimates are correlated by virtue of their dependence on the gas-exchange formalism. This was taken into account in producing the global total but is not otherwise reflected in the constraints given above. However Figure 7 shows that any further constraint on the southern ocean sink would constrain the global total.

Because of the approximate nature of the transport relation used to derive the sources, it is inappropriate to assign great significance to the particular numbers. The important point to note is that even though the uncertainties in source estimates from transport modelling may be quite large, such information can still give useful constraints on the atmospheric carbon budget.

6. Concluding remarks

The preceding sections have described the linear programming formalism for analysing atmospheric carbon budgets and have presented some preliminary examples of the application of this approach. The analyses have been presented at three levels of detail, chosen (somewhat arbitrarily) as being based on:

- Direct estimates of the source components.
- Direct estimates supplemented by global modelling of oceanic uptake.
- Combinations of regional budgets, related using atmospheric transport modelling.

These studies have revealed some of the key weaknesses in our current knowledge.

In order to reduce the uncertainties in the atmospheric carbon budget there are two approaches (by no means mutually exclusive). The first is to improve the precision of the estimates used. The second is to incorporate new types of information.



Figure 6: Joint variation of CLEARS and T_CLER, given the constraints from Table 2 for the 1980–5 case and the regional constraints listed in equations (7) to (15).



Figure 7: Joint variation of total ocean sink, OCEANS, and southern oceans sink, S_SEAS, given the constraints from Table 2 for the 1980–5 case and the regional constraints listed in equations (7) to (15).

The areas in which there may be scope for improvement are:

- Regional source budgets from atmospheric transport modelling. The construction of error estimates is a difficult problem but one that is being addressed. One of the motivations for the preliminary analysis presented in Section 5 was to explore the requirements for incorporating regional information into overall atmospheric budgets. The results show that even though the ill-conditioned nature of the source deduction problem leads to large uncertainties in regional source distributions, such regional information can still give useful constraints on the global budget.
- The area in which improvements in the direct estimates is most desirable is in the 'fertilisation' contribution. Present estimates a based on extrapolation of laboratory-determined β values to the global scale. Although a number of model calculations have been made, they are subject to very considerable uncertainty and there seems to be little hope of refining the precision to the extent that would give independent constraints on the atmospheric carbon budget.
- There would seem to be a strong possibility that uncertainties in direct estimates of air-sea fluxes will be reduced, at least by the extent implied by uncertainties in the gas-exchange coefficient. Direct flux measurements should be able to distinguish between the two forms and hopefully lead to an understanding of how the discrepancy has arisen. This would provide a significant reduction in the uncertainties in direct estimates of oceanic CO₂ uptake. Further improvements could be obtained (with considerable effort) from more comprehensive sampling of p_{CO2} . However, the best hope for determining the oceanic CO₂ uptake may still lie with ocean GCMs.

Some of the additional types of information that could be incorporated into atmospheric carbon budgeting studies are:

- Data on ¹³C distributions. In particular regional data should be able to distinguish oceanic from biotic contributions. Forward modelling of the ¹³C distribution has been presented by Pearman and Hyson (1986) and Keeling et al. (1989b). What is needed is to quantify the uncertainties in such analyses in the same ways as is needed for CO₂.
- As noted above, ocean GCMs may be able to give additional information about the regional distribution of CO₂ uptake.

This report has a number of objectives:

- To present initial budget estimates to provide base cases against which to measure improvements in our knowledge;
- To guide modelling studies that seek to explore the range of uncertainty in the atmospheric carbon budget;
- To present a formalism that can be applied to other species such as methane.

Disclaimer

The budgets presented here are preliminary descriptions based on information available to the author. They are presented to illustrate two key points: the need for recognising the secular change when analysing atmospheric carbon budgets and the need to consider the error structure, particularly the error covariance when using regional budgets derived from atmospheric transport modelling.

These budgets are subject to several caveats:

- It has not always been possible to compare like with like in making some of the crosscomparisons. In particular the error bounds are taken from (or derived from) published sources. In some cases the use that is made above may imply a higher precision than the original authors intended. It is hoped that by describing the types of information required for carbon budgeting studies, this report will encourage the presentation of analyses of individual subsytems in a way that facilitates using the results in carbon budgeting studies.
- The linear programming technique used here gives a fitting procedure related to fitting under a ℓ_{∞} norm. Such fitting is very sensitive to 'outliers' i.e. information that contains 'error' contributions inconsistent with the underlying system being analysed.
- The consistency regions determined from the regional budgets are purely illustrative, partly because the ranges in direct estimates were not always readily available but mainly because the techniques for producing the requisite information from atmospheric transport modelling are poorly developed. The analysis in Section 5 shows the type of error information that is required so that future transport modelling can address the needs of carbon budgeting studies.

Acknowledgements

The author wishes to thank Prof. Paul Quay for valuable comments on the manuscript. Studies of the atmospheric carbon budget and the role of atmospheric transport modelling in reducing these uncertainties are carried out in support of the Division of Atmospheric Research threedimensional transport modelling project, supported by the State Electricity Commission of Victoria. Thanks are due to Ms. P.V. Giuffré-Enting for supplying computing equipment used in some of the development of the programs used in the analysis presented here.

References

- Bacastow, R.B. and Maier-Reimer, E. 1990. Ocean circulation model of the carbon cycle. *Climate Dynamics*, **4**, 95–125.
- Bolin, B., Björkström, A., Holmén, K. and Moore, B. 1983. The simultaneous use of tracers for ocean circulation studies. *Tellus*, **35B**, 206–236.
- Broecker, W.S., Peng, T.-H. and Engh, R. 1980. Modeling the carbon system. *Radiocarbon*, **22**, 565–598.
- CDIAC 1990. *Trends '90: A Compendium of Data on Global Change*. Ed. T.A. Boden, P. Kanciruk and M.P. Farrell. Carbon Dioxide Information and Analysis Center. (Oak Ridge, Tennessee).
- CDIAC 1991. *Trends '91: A Compendium of Data on Global Change*. Ed. T.A. Boden, R.J. Sepanski and F.W. Stoss. Carbon Dioxide Information and Analysis Center. (Oak Ridge, Tennessee).
- CMDL 1990. *Climate Monitoring and Diagnostics Laboratory. No. 18. Summary Report 1989.* Ed. W. D. Komhyr and. R.M. Rosson. (US Dept. Commerce).
- Enting, I.G. 1985. Principles of constrained inversion in the calibration of carbon cycle models. *Tellus*, **37B**, 7–27.
- Enting, I.G. 1987. The interannual variation in the seasonal cycle of carbon dioxide concentration at Mauna Loa. J. Geophys. Res., **92D**, 5497–5504.
- Enting, I.G. 1990. Ambiguities in the calibration of carbon cycle models. *Inverse Problems*, 6, L39–L46.
- Enting, I.G. 1991. *Calculating Future Atmospheric CO*₂ *Concentrations*. Division of Atmospheric Research, Technical Paper 22. (CSIRO).
- Enting, I.G. 1992. The incompatibility of ice-core CO_2 data with reconstructions of biotic CO_2 sources. II. The influence of CO_2 -fertilised growth. *Tellus*, **44B**, 23–32.
- Enting, I.G. and Manning, M.R. 1989. The seasonal cycle of CO_2 at Mauna Loa: a re-analysis of a digital filtering study. pp124–131 of *The Statistical Treatment of CO*₂ *Data Records*. Ed. W.P. Elliott. NOAA Technical Memorandum ERL ARL-173.
- Enting, I.G. and Mansbridge, J.V. 1987. The incompatibility of ice-core CO₂ data with reconstructions of biotic CO₂ sources. *Tellus*, **39B**, 318–325.
- Enting, I.G. and Mansbridge, J.V. 1987. Inversion relations for the deconvolution of CO₂ data from ice cores. *Inverse Problems*, **3**, L63–L69.
- Enting, I.G. and Mansbridge, J.V. 1989. Seasonal sources and sinks of atmospheric CO₂: Direct inversion of filtered data. *Tellus*, **41B**, 111–126.
- Enting, I.G. and Mansbridge, J.V. 1991. Latitudinal distribution of sources and sinks of CO₂: Results of an inversion study. *Tellus*, **43B**, 156–170.
- Enting, I.G. and Newsam, G.N. 1990. Inverse problems in atmospheric constituent studies: II. Sources in the free atmosphere. *Inverse Problems*, **6**, 349–362.
- Enting, I.G. and Pearman, G.I. 1986. The use of observations in calibrating and validating carbon cycle models. in *The Changing Carbon Cycle: A Global Analysis*. Ed. J. Trabalka and D. Reichle. (Springer-Verlag, New York).
- Enting, I.G. and Pearman, G.I. 1987. Description of a one-dimensional carbon cycle model calibrated using techniques of constrained inversion. *Tellus*, **39B**, 459–476.
- Enting, I.G. and Pearman, G.I. 1992. Average global distribution of CO₂. In *The Global Carbon Cycle: Proceedings of the NATO Advanced Study Institute, Il Ciocco, Italy, September 1991.*

Ed. M. Heimann. (Springer-Verlag). (In press).

- Fung, I., Prentice, K., Matthews, E., Lerner, J. and Russell, G. 1983. Three-dimensional tracer model study of atmospheric CO₂: response to seasonal exchanges with the terrestrial biosphere. *J.Geophys. Res.*, 88C, 1281–1294.
- Gardner, R.H. and Trabalka, J.R. 1985. *Methods of Uncertainty Analysis for a Global Carbon Cycle Model*. Technical Report TR-024. (US Dept. of Energy).
- Gaudry, A., Monfray, P., Polian, G. and Lambert, G. 1978. The 1982–1983 El Niño: a 6 billion ton CO₂ release. *Tellus*, **39B**, 209–213.
- Houghton, R.A., Hobbie, J.E., Melillo, J.M., Moore, B., Peterson, B.J., Shaver, G.R. and Woodwell, G.M. 1983. Changes in the carbon content of terrestrial biota and soils between 1860 and 1980: A net release of CO₂ to the atmosphere. *Ecol. Monog.*, **53**, 235–262.
- Houghton, R.A., Boone, R.D., Fruci, J.R., Hobbie, J.E., Melillo, J.M., Palm, C.A., Peterson, B.J., Shaver, G.R., Woodwell, G.M., Moore, B.M., Skole, D.L., and Myers, N. 1987. The flux of carbon dioxide from tropical ecosystems to the atmosphere in 1980 due to changes in land use: geographic distribution in the global flux. *Tellus*, **39B**, 122–139.
- Idso, S.B. 1987. Detection of global carbon dioxide effects. Nature 329, 293.
- IPCC 1990. *Climate Change: The IPCC Scientific Assessment*. Ed. J.T. Houghton, G.J. Jenkins and J.J. Ephraums. Published for the InterGovernmental Panel on Climate Change by Cambridge University Press.
- Keeling, C.D. 1973. Industrial production of carbon dioxide from fossil fuels and limestone. *Tellus 25*, 174–198.
- Keeling, C.D., Bacastow, R.B., Carter, A.F., Piper, S.C., Whorf, T.P., Heimann, M., Mook, W.G. and Roeloffzen, H. 1989a. A three-dimensional model of atmospheric transport based on observed winds: 1. Analysis of observational data. pp 165–236 of *Aspects of Climate Variability in the Pacific and Western Americas*. Ed. D.H. Peterson. AGU Monograph 55. (AGU, Washington).
- Keeling, C.D., Piper, S.C. and Heimann, M. 1989b. A three-dimensional model of atmospheric transport based on observed winds: 4. Mean annual gradients and interannual variations. pp 305–363 of *Aspects of Climate Variability in the Pacific and Western Americas*. Ed. D.H. Peterson. AGU Monograph 55. (AGU, Washington).
- Kohlmaier, G.H., Bröhl, H., Siré, E.O., Plöchl, M. and Revelle, R. 1987. Modelling stimulation of plants and ecosysytem response to present levels of atmospheric CO₂. *Tellus*, **39B**, 155-170.
- La Marche, V.C., Graybill, D.A., Fritts, H.C., and Rose, M.R. 1984. Increasing atmospheric carbon dioxide: tree ring evidence for growth enhancement in natural vegetation. *Science*, 225, 1019–1021.
- Liss, P.S. and Merlivat, L. 1986. Air-sea gas exchange rates: Introduction and synthesis. pp113– 127 of *The Role of Air-Sea Exchange in Geochemical Cycling*. Ed. P. Buat-Menard. (Reidel, Hingham).
- Maier-Reimer, E. and Hasselmann, K. 1987. Transport and storage of CO_2 in the ocean an inorganic ocean-circulation carbon cycle model. *Climate Dynamics*, **2**, 63–90.
- Marland, G. and Rotty, R.M. 1984. Carbon dioxide emissions from fossil fuels: A procedure for estimation and results for 1950–1982. *Tellus*, **36B**, 232–261.
- Moore, B. and Björkström, A. 1986. Calibrating Ocean models by the constrainted inverse method. pp 295–328 of *The Changing Carbon Cycle: A Global Analysis*. Ed. J. Trabalka and D. Reichle. (Springer-Verlag, New York).

- Neftel, A., Moor, E., Oeschger, H. and Stauffer, B. 1985. Evidence from polar ice cores for the increase in atmospheric CO₂ in the past two centuries. *Nature*, **315**, 45–47.
- Newsam, G.N. and Enting, I.G. 1988

. Inverse problems in atmospheric constituent studies: I. Determination of surface sources under a diffusive transport approximation. *Inverse Problems*, **4**, 1037–1054.

- Oeschger, H. and Heimann, M. 1983. Uncertainties of predictions of future atmospheric CO₂ concentrations. *J. Geophys. Res*, **88C**, 1258–1262.
- Oeschger, H., Siegenthaler, U., Schotterer, U. and Gugelmann, A. 1975. A box diffusion model to study the carbon dioxide exchange in nature. *Tellus*, **27**, 168–192.
- Pearman, G.I. and Hyson, P. 1980. Activities of the global biosphere reflected in atmospheric CO₂ records. *J. Geophys. Res.*, **85**, 4457–4467.
- Pearman, G.I. and Hyson, P. 1986. Global transport and inter-reservoir exchange of carbon dioxide with particular reference to stable isotope distribution. *J. Atmos. Chem.*, **4**, 81–124.
- Pearman, G.I., Hyson, P. and Fraser, P.J. 1983. The global distribution of atmospheric carbon dioxide: 1. Aspects of observations and modelling. *J. Geophys. Res.*, **88C**, 3581–3590.
- Press, W.H., Flannery, B.P., Teukolsky, S.A. and Vetterling, W.T. 1986. *Numerical Recipes: The Art of Scientific Computing*. (Cambridge University Press, Cambridge).
- Rotty, R.M. 1987. A look at 1983 CO₂ emissions from fossil fuels (with preliminary data for 1984). *Tellus*, **39B**, 203–208.
- Sarmiento, J.L. 1986. Three-dimensional ocean models for predicting the distribution of CO₂ between the ocean and atmosphere. pp 277–294 of *The Changing Carbon Cycle: A Global Analysis*. Ed. J. Trabalka and D. Reichle. (Springer-Verlag, New York).
- Sarmiento, J.L. and Sundquist, E. 1992. *preprint entitled:* Oceanic uptake of anthropogenic CO₂: A new budget.
- Siegenthaler, U. and Oeschger, H. 1987. Biospheric CO₂ emissions during the past 200 years reconstructed by deconvolution of ice core data. *Tellus*, **39B**, 140–154.
- Tans, P.P., Conway, T.J. and Nakazawa, T. 1989. Latitudinal distribution of the sources and sinks of atmospheric carbon dioxide derived from surface observations and an atmospheric transport model. J. Geophys. Res., 94D, 5151–5172.
- Tans, P.P., Fung, I.Y. and Takahashi, T. 1990. Observational constraints on the global atmospheric CO₂ budget. *Science*, **247**, 1431–1438.
- Thompson, M.L., Enting, I.G., Pearman, G.I. and Hyson, P. 1986. Interannual variation of atmospheric CO₂ concentration. *J. Atmos. Chem.*, **4**, 125–155.

CSIRO Atmospheric Research Technical Papers

This series has been issued as *Division of Atmospheric Research Technical Paper* (nos. 1–19); *CSIRO Division of Atmospheric Research Technical Paper* (nos. 20–37) and *CSIRO Atmospheric Research Technical Paper* from no. 38.

Regular electronic publication commenced with no. 45. Earlier technical papers are progressively being made available in electronic form. A current list of technical papers is maintained at *http://www.dar.csiro.au/info/TP.htm.* Papers may be issued out of sequence.

- No. 1 Galbally, I.E.; Roy, C.R.; O'Brien, R.S.; Ridley, B.A.; Hastie, D.R.; Evans, W.J.F.;
 McElroy, C.T.; Kerr, J.B.; Hyson, P.; Knight, W.; Laby, J.E. *Measurements of trace composition of the Austral stratosphere: chemical and meteorological data*. 1983. 31
 p.
- No. 2 Enting, I.G. *Error analysis for parameter estimates from constrained inversion*. 1983. 18 p.
- No. 3 Enting, I.G.; Pearman, G.I. *Refinements to a one-dimensional carbon cycle model*. 1983. 35 p.
- No. 4 Francey, R.J.; Barbetti, M.; Bird, T.; Beardsmore, D.; Coupland, W.; Dolezal, J.E.;
 Farquhar, G.D.; Flynn, R.G.; Fraser, P.J.; Gifford, R.M.; Goodman, H.S.; Kunda, B.;
 McPhail, S.; Nanson, G.; Pearman, G.I.; Richards, N.G.; Sharkey, T.D.; Temple, R.B.;
 Weir, B. *Isotopes in tree rings.* 1984. 86 p.
- No. 5 Enting, I.G. Techniques for determining surface sources from surface observations of *atmospheric constituents*. 1984. 30 p.
- No. 6 Beardsmore, D.J.; Pearman, G.I.; O'Brien, R.C. *The CSIRO (Australia) Atmospheric Carbon Dioxide Monitoring Program: surface data*. 1984. 115 p.
- No. 7 Scott, John C. *High speed magnetic tape interface for a microcomputer*. 1984. 17 p.
- No. 8 Galbally, I.E.; Roy, C.R.; Elsworth, C.M.; Rabich, H.A.H. *The measurement of nitrogen oxide* (*NO*, *NO*₂) *exchange over plant/soil surfaces*. 1985. 23 p.
- No. 9 Enting, I.G. A strategy for calibrating atmospheric transport models. 1985. 25 p.
- No. 10 O'Brien, D.M. TOVPIX: software for extraction and calibration of TOVS data from the high resolution picture transmission from TIROS-N satellites. 1985. 41 p.
- No. 11 Enting, I.G.; Mansbridge, J.V. Description of a two-dimensional atmospheric transport model. 1986. 22 p.
- No. 12 Everett, J.R.; O'Brien, D.M.; Davis, T.J. A report on experiments to measure average fibre diameters by optical fourier analysis. 1986. 22 p.
- No. 13 Enting, I.G. A signal processing approach to analysing background atmospheric constituent data. 1986. 21 p.
- No. 14 Enting, I.G.; Mansbridge, J.V. Preliminary studies with a two-dimensional model using transport fields derived from a GCM. 1987. 47 p.
- No. 15 O'Brien, D.M.; Mitchell, R.M. Technical assessment of the joint CSIRO/Bureau of Meteorology proposal for a geostationary imager/sounder over the Australian region. 1987. 53 p.
- No. 16 Galbally, I.E.; Manins, P.C.; Ripari, L.; Bateup, R. A numerical model of the late (ascending) stage of a nuclear fireball. 1987. 89 p.

- No. 17 Durre, A.M.; Beer, T. Wind information prediction study: Annaburroo meteorological data analysis. 1989. 30 p. + diskette.
- No. 18 Mansbridge, J.V.; Enting, I.G. Sensitivity studies in a two-dimensional atmospheric transport model. 1989. 33 p.
- No. 19 O'Brien, D.M.; Mitchell, R.M. Zones of feasibility for retrieval of surface pressure from observations of absorption in the A band of oxygen. 1989. 12 p.
- No. 20 Evans, J.L. Envisaged impacts of enhanced greenhouse warming on tropical cyclones in the Australian region. 1990. 31 p. [Out of print]
- No. 21 Whetton, P.H.; Pittock, A.B. Australian region intercomparison of the results of some general circulation models used in enhanced greenhouse experiments. 1991. 73 p. [Out of print]
- No. 22 Enting, I.G. Calculating future atmospheric CO₂ concentrations. 1991. 32 p. Also electronic edition (718 kB pdf), at http://www.dar.csiro.au/publications/Enting_2000d.pdf
- No. 23 Kowalczyk, E.A.; Garratt, J.R.; Krummel, P.B. A soil-canopy scheme for use in a numerical model of the atmosphere 1D stand-alone model. 1992. 56 p.
- No. 24 Physick, W.L.; Noonan, J.A.; McGregor, J.L.; Hurley, P.J.; Abbs, D.J.; Manins, P.C. *LADM: A Lagrangian Atmospheric Dispersion Model*. 1994. 137 p.
- No. 25 Enting, I.G. *Constraining the atmospheric carbon budget: a preliminary assessment.* 1992. 28 p. Also electronic edition (571 kB), at http://www.dar.csiro.au/publications/Enting_2000b.pdf
- No. 26 McGregor, J.L.; Gordon, H.B.; Watterson, I.G.; Dix, M.R.; Rotstayn, L.D. *The CSIRO* 9-level atmospheric general circulation model. 1993. 89 p.
- No. 27 Enting, I.G.; Lassey, K.R. *Projections of future CO*₂. with appendix by R.A. Houghton. 1993. 42 p. Also electronic edition (860 kB pdf), at *http://www.dar.csiro.au/publications/Enting_2000e.pdf*
- No. 28 [Not published]
- No. 29 Enting, I.G.; Trudinger, C.M.; Francey, R.J.; Granek, H. Synthesis inversion of atmospheric CO₂ using the GISS tracer transport model. 1993. 44 p.
- No. 30 O'Brien, D.M. Radiation fluxes and cloud amounts predicted by the CSIRO nine level GCM and observed by ERBE and ISCCP. 1993. 37 p.
- No. 31 Enting, I.G.; Wigley, T.M.L.; Heimann, M. Future emissions and concentrations of carbon dioxide: key ocean/atmosphere/land analyses. 1993. 120 p.
- No. 32 Kowalczyk, E.A.; Garratt, J.R.; Krummel, P.B. *Implementation of a soil-canopy scheme into the CSIRO GCM – regional aspects of the model response*. 1994. 59 p.
- No. 33 Prata, A.J. Validation data for land surface temperature determination from satellites. 1994. 36 p.
- No. 34 Dilley, A.C.; Elsum, C.C. *Improved AVHRR data navigation using automated land feature recognition to correct a satellite orbital model.* 1994. 22 p.
- No. 35 Hill, R.H.; Long, A.B. The CSIRO dual-frequency microwave radiometer. 1995. 16 p.
- No. 36 Rayner, P.J.; Law, R.M. A comparison of modelled responses to prescribed CO₂ sources. 1995. 84 p.
- No. 37 Hennessy, K.J. CSIRO Climate change output. 1998. 23 p.
- No. 38 Enting, I.G. Attribution of greenhouse gas emissions, concentrations and radiative forcing. 1998. 27 p. Also electronic edition (557 kB pdf), at http://www.dar.csiro.au/publications/Enting_2000c.pdf

- No. 39 O'Brien, D.M.; Tregoning, P. Geographical distributions of occultations of GPS satellites viewed from a low earth orbiting satellite. (1998) 28p.
- No. 40 Enting, I.G. Characterising the temporal variability of the global carbon cycle. 1999.
 53 p. Also electronic edition (636 kB pdf), at http://www.dar.csiro.au/publications/Enting_2000a.pdf
- No. 41 (in preparation).
- No. 42 Mitchell, R.M. Calibration status of the NOAA AVHRR solar reflectance channels: CalWatch revision 1. 1999. 20 p.
- No. 43 Hurley, P.J. *The Air Pollution Model (TAPM) Version 1: technical description and examples.* 1999. 41 p. Also electronic edition (276 kB pdf), at http://www.dar.csiro.au/publications/Hurley_1999a.pdf
- No. 44 Frederiksen, J.S.; Dix, M.R.; Davies, A.G. A new eddy diffusion parameterisation for the CSIRO GCM. 2000. 31 p.
- No. 45 Young, S.A. Vegetation Lidar Studies. Electronic edition in preparation.
- No. 46 Prata, A. J. Global Distribution of Maximum Land Surface Temperature Inferred from Satellites: Implications for the Operation of the Advanced Along Track Scanning Radiometer. 2000. 30 p. Electronic edition only (8307 kB pdf), at http://www.dar.csiro.au/publications/Prata_2000a.pdf
- No. 47 Prata, A. J. Precipitable water retrieval from multi-filter rotating shadowband radiometer measurements. 2000. 14 p. Electronic edition only (1554 kB pdf), at http://www.dar.csiro.au/publications/Prata_2000b.pdf
- No. 48 Prata, A.J. and Grant, I.F. *Determination of mass loadings and plume heights of volcanic ash clouds from satellite data.* (in preparation)
- No. 49 (in preparation)
- No. 50 (in preparation)
- No. 51 Meyer, C.P., Galbally, I.E., Wang, Y.-P., Weeks, I.A., Jamie, I. and Griffith, D.W.T. *Two automatic chamber techniques for measuring soil-atmosphere exchanges of trace gases and results of their use in the OASIS field experiment.* Electronic edition only (pdf),

at http://www.dar.csiro.au/publications/Meyer_2001a.pdf

Address and contact details: CSIRO Atmospheric Research Private Bag No.1 Aspendale Victoria 3195 Australia Ph: (+61 3) 9239 4400; fax: (+61 3) 9239 4444 e-mail: chief@dar.csiro.au

Last updated 27 March 2001